Author's response to reviews

Title: Clinical Utility of a Nested Nucleic Acid Amplification Format in Comparison to Viral Culture for the Diagnosis of Mucosal Herpes Simplex Infection

Authors:

Peter V Coyle (peter.coyle@bll.n-i.nhs.uk)
Hugh J O'Neill (hugh.oneill@bll.n-i.nhs.uk)
Conall McCaughey (conall.mccaughey@bll.n-i.nhs.uk)
Dorothy E Wyatt (dorothy.wyatt@bll.n-i.nhs.uk)
Michael O McBride (michael.mcbride@royalhospitals.n-i.nhs.uk)

Version: 2 Date: 20 Aug 2001

20/08/2001
BioMed Central Editorial Team,

Re: Clinical Utility of a Nested Nucleic Acid Amplification Format in Comparison to Viral Culture for the Diagnosis of Mucosal Herpes Simplex Infection

I am submitting the final version of this paper for consideration for publication in BioMed Central. I have made some final corrections in the layout to the paper in line with the checklist provided and have reduced the abstract to 250 words. The paper contains the corrections previously inserted after consideration of the reviewers' comments but no additional changes. I will e-mail the signed checklist as requested.

Yours sincerely,
Dr. P.V. Coyle
Consultant Virologist.